

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL103294

Key Points:

- Small changes in water and clay content of the bed significantly affect debris-flow erosion processes and magnitude
- Bed-water content increases erosion when the bed is nearly saturated, whereas for clay content an optimum exists for erosion around 3%-4%
- Water and clay content of the bed affect debris-flow erosion by affecting bed pore pressure when the debris flow overrides the bed

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

L. Roelofs, l.roelofs@uu.nl

Citation:

Roelofs, L., Nota, E. W., Flipsen, T. C. W., Colucci, P., & de Haas, T. (2023). How bed composition affects erosion by debris flows—An experimental assessment. *Geophysical Research Letters*, 50, e2023GL103294. https://doi. org/10.1029/2023GL103294

Received 14 FEB 2023 Accepted 27 JUN 2023

Author Contributions:

Conceptualization: Lonneke Roelofs, Eise W. Nota, Tom C. W. Flipsen, Pauline Colucci, Tjalling de Haas Data curation: Lonneke Roelofs, Eise W. Nota, Tom C. W. Flipsen, Pauline Colucci Formal analysis: Lonneke Roelofs Funding acquisition: Tjalling de Haas Investigation: Lonneke Roelofs, Eise W. Nota, Tom C. W. Flipsen, Pauline Colucci Methodology: Lonneke Roelofs, Tjalling de Haas Project Administration: Tjalling de Haas Resources: Tjalling de Haas

Supervision: Lonneke Roelofs, Tjalling de Haas

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

How Bed Composition Affects Erosion by Debris Flows—An Experimental Assessment

Lonneke Roelofs¹, Eise W. Nota¹, Tom C. W. Flipsen¹, Pauline Colucci¹, and Tjalling de Haas¹

¹Department of Physical Geography, Faculty of Geosciences, Utrecht University, Utrecht, The Netherlands

Abstract A solid physical understanding of debris-flow erosion is needed for both hazard prediction and understanding landscape evolution. However, the processes and forces involved in erosion by debris flows and especially how the erodible surface itself influences erosion are poorly understood. Here, we experimentally investigate the effects of bed composition on debris-flow erosion, by systematically varying the composition of an erodible bed in a small-scale debris-flow flume. The experiments show that water and clay content of an unconsolidated bed significantly control erosion magnitude by affecting the transfer of pore pressure, loading conditions, and contraction-dilation behavior of the bed. As the water content increases and the bed comes close to saturation, erosion increases rapidly, whereas for clay content an optimum for erosion exists around a clay content of 3%–4%. Our results show that small variations in bed composition can have large effects on debris-flow erosion, and thus volume growth and hazard potential.

Plain Language Summary Debris flows are slurries of water, soil and rock that rush down mountainsides. In their path down-slope they erode material and ultimately may build up depositional fans in lower laying areas. These fans are often preferred sites for settlement. This means that new debris flows directly threaten human life and infrastructure. We know from other studies that the bigger the debris flow the larger the number of casualties. And we also know that debris flows rapidly increase in size when rushing down the mountain, by eroding and picking up loose sediment and rock. However, current computer models, used for hazard prediction, are bad at predicting erosion and therefore debris-flow size. We believe that an important factor for debris-flow erosion is overlooked in these models; the composition of the eroded material. Our experiments with small debris flows in the lab show that the amount of clay and water in the soil control how much erosion occurs. More water in the bed increases erosion, and for clay content, an optimum exists for maximum erosion. This eventually implies that the geology of the catchment and the soil moisture conditions should be assessed carefully when making predictions on debris-flow hazard.

1. Introduction

Debris flows are an active geomorphological agent that, on the short term, pose a threat to human life, property and infrastructure (e.g., Beguería et al., 2009; Dowling & Santi, 2014; Luna et al., 2012; Rickenmann, 1999, 2005; Zou et al., 2020). In the long term, debris flows play an important role in landscape evolution by eroding soil and rock, cutting valleys, and depositing sediments in large fan systems (Blair & McPherson, 1994; Cavalli & Marchi, 2008; De Haas et al., 2014; de Haas et al., 2018; Stock & Dietrich, 2003, 2006). Understanding debrisflow erosion is important to explain long-term landscape evolution, but it is also crucial for mitigating risks posed by debris flows. Debris-flow volume has directly been linked to the number of casualties (Dowling & Santi, 2014), and volume growth of the debris flow, due to erosion and sediment entrainment, can be several orders of magnitude larger than the initial flow volume (e.g., Frank et al., 2015; Hungr et al., 2005; Navratil et al., 2013; Santi et al., 2008; Simoni et al., 2020; Takahashi, 1978). In current hazard analyses, volume growth is often predicted based on the availability of loose sediment (e.g., De Haas et al., 2020; Jakob, 2005), on volumes of past debris flows (e.g., Conway et al., 2010; de Haas et al., 2022; Ékes & Friele, 2003; Giraud, 2005), catchment and watershed characteristics (e.g., de Haas & Densmore, 2019; Takahashi, 1981; Wan & Lei, 2009; Welsh & Davies, 2011; Wilford et al., 2004) or on linear regression between peak discharge and volume (e.g., Rickenmann, 1999). These criteria are based on the intrinsic and autogenic settings of the debris-flow systems. However, when boundary conditions change, for example, by anthropogenic impacts or climate change, debrisflow hazard will change accordingly (Cannon & DeGraff, 2009; Lugon & Stoffel, 2010; Rebetez et al., 1997; Stoffel et al., 2014; Stoffel & Huggel, 2012) and hazard predictions based on intrinsic settings will not always suffice.



Visualization: Lonneke Roelofs Writing – original draft: Lonneke Roelofs Writing – review & editing: Lonneke Roelofs, Eise W. Nota, Tom C. W. Flipsen, Pauline Colucci, Tjalling de Haas The importance of erosion in debris-flow volume growth has led to an increase in the number of numerical debrisflow models that incorporate erosion (Abancó & Hürlimann, 2014; Baggio et al., 2021; Frank et al., 2015; Han et al., 2016; Hungr et al., 2005; Iverson, 2012; Iverson & Ouyang, 2015; Li et al., 2020; Takahashi, 1978). However, the vast amount of approaches, from empirical to physics-based, and the varying incorporated physical mechanisms, highlight the lack of a unified debris-flow erosion theory (De Haas et al., 2020) and the need for a better physical understanding of the involved processes and parameters. Experiments in large- and small-scale flumes have highlighted the importance of certain parameters on erosion processes. For example, De Haas and Woerkom (2016) and Roelofs et al. (2022) showed that the water, gravel, and clay content of the debris flow itself affect erosion magnitude and patterns by changing the erosional shear and impact forces as well as the pore pressure in the debris flow that is transferred to the top layer of the bed. These studies showed that increasing gravel and water content linearly relates to an increase in erosion (De Haas & Woerkom, 2016; Roelofs et al., 2022), whereas clay content non-linearly interacts with erosion via interstitial fluid viscosity and increased pore pressures (Roelofs et al., 2022). In addition, experiments from large-scale flume studies by Iverson et al. (2011) and Reid et al. (2011) show that higher water content of the bed results in higher bed pore pressures and larger quantities of erosion. This finding is in line with observations from the field (de Haas et al., 2022; McCoy et al., 2012) and the long-standing theory on how increased pore pressures facilitate erosion by debris flows (Bagnold, 1954; Hungr et al., 2005; Iverson, 1997; Li et al., 2020; McCoy et al., 2012).

The large influence of water content of the bed on erosion can be explained by the difference between undrained and drained loading conditions. Drained loading occurs when air and fluid are able to drain through the pores without increasing pore pressure. In contrast, under undrained loading conditions, pore fluid in the soil is unable to drain out or into the pores, leading to increased or decreased pore pressure. In the most extreme case, the pore fluid bears the entire unit weight of the saturated debris (Major, 2000). Increased pore pressure decreases intergranular friction between the grains and can enable liquefaction of the sediment, which enhances the erodibility of the bed (Hungr et al., 2005; Iverson, 2012; Major, 2000; Sassa & hui Wang, 2005). The dissipation of excess pore pressure can be described by the hydraulic diffusivity *D* (Major, 2000; McCoy et al., 2012). Diffusivity is controlled by the water content of the bed but also by the characteristics of the soil, that is, permeability and matrix compressibility (see Major (2000) and McCoy et al. (2012) for mathematical description). These are, themselves defined by the grain-size distribution of the soil and the clay content. In addition, the content of fines in the bed also influences the dynamic viscosity of the interstitial fluid and the fluid compressibility as it penetrates into the bed.

Another physical soil characteristic that influences erosion susceptibility is shear strength θ , which is the ability of soils to resist movement along a slip surface. Shear strength is dependent on the composition of the soil, the level of compaction, and moisture content. Furthermore, it is important to consider how soils react to compaction and shearing, as this directly influences pore pressures in the bed. When soils dilate in response to shear, pore pressures decrease, and when soils contract in response to shearing, pore pressures increase (Iverson, 2012; Iverson & Ouyang, 2015; McCoy et al., 2012). Modeling work by Iverson (2012) shows that if finer sediment is present in a bed, overridden by a debris flow, slight shear displacement can play a dominant role in generating pore fluid pressures. When enough fine sediment is present this makes the soil behave effectively undrained (Iverson, 2012). However, to date, it remains unclear how these balancing forces and processes influence erosion by debris flows for different soil compositions. To advance our understanding of debris-flow erosion, we want to elucidate the effects of bed composition on debris-flow erosion processes. We aim to determine and quantify the effects of clay and water content of the bed on debris-flow erosion processes and magnitude. We also aim at gaining a better understanding of the interaction between different erosion mechanisms and influencing parameters, for example, liquefaction, drained versus undrained loading, and diffusivity. To this end, we perform experiments in a small-scale debris-flow flume with an erodible bed to systematically test the influence of the bed's clay and water content on erosion processes and magnitude.

2. Materials and Methods

2.1. Flume Set-Up, Bed Composition and Data-Analyses

To study and quantify the effects of bed composition on debris-flow erosion magnitude and processes, we combined a series of flume experiments with geotechnical tests to determine the diffusivity and porosity of the soils used in the flume. The flume consisted of a 5.4 m long and 0.3 m wide chute with a depression in the lower 2.5 m, a mixing tank with a forced-action mixer (Baron E120), and a custom-made release gate (See Figure 1) (set-up is similar to de Haas et al. (2021) and Roelofs et al. (2022)). In the depression in the lower half of the chute,



10.1029/2023GL103294



Figure 1. Schematic (a) and photo (b) of the flume. Sketch: orange rectangle represents the force action mixer and the dotted yellow trapezoid represents the erodible bed. All dimensions are in centimeters.

an erodible, unsaturated, and loosely packed bed was created, of which we systematically varied the composition. Erosion under the different bed compositions was tested at three different flume angles: 28° , 31° , 34° . For all experiments, the debris-flow composition was kept constant (see Table S1 in Supporting Information S1). To ensure repeatability and account for natural variability, every experimental setting was performed twice. Within the debris flows, frictional forces dominated flow dynamics (see Table S1 in Supporting Information S1), similar to most debris flows in nature (Zhou & Ng, 2010). For a small number of experiments, conducted under a flume angle of 28° , viscous forces dominated over collisional forces.

The loosely packed bed consisted of sand, clay (kaolin), and water in different ratios. To test the influence of the water content of the bed, we used a sandy bed without clay in which we systematically varied the total mass fraction of water from 0.09 to 0.15 (Table S1 in Supporting Information S1). To test the influence of clay content of the bed, we used a total mass fraction of water of 0.11 and varied the dry mass fraction of clay within the sandy bed from 0 to 0.1, while keeping the sand porosity roughly constant (Table S1 in Supporting Information S1). For every experiment, the bed was prepared by mixing the sediment and water with a hand-held mortar mixer, after which the mixture was placed in the recess in the lower half of the flume. For the grain-size distribution of the used sediments see Figure S1 in Supporting Information S1.

Two pore pressure sensors were installed underneath the flume, 50 cm downslope of the start of the erodible bed. These sensors were connected to small plastic tubes, with permeable filters, that protruded into the bed at different heights (3 and 4 cm below the surface). The small tubes above the sensors were filled with de-aired water before every experiment and the recorded hydrostatic pressure was used as the reference pressure.

To quantify the net erosion, the bed was scanned using a Vialux z-Snapper 3D scanner. This scanner created a 3D point cloud of sub-mm accuracy of the bed by structured light and imaging before and after the debris flow had passed. The point clouds were denoised and transformed into gridded digital elevation models (DEMs) of 0.3 mm resolution by natural neighbor interpolation.

2.2. Diffusivity and Porosity Tests

The diffusivity of the bed compositions with varying amounts of clay was determined following the methodology of Major (2000). Tests were conducted for dry bulk clay mass fractions ranging from 0 to 0.1, with a step size

of 0.01. A smooth transparent tube with a radius of 6.25 cm was filled with a sand-clay-water mixture up to an approximate height of 55 cm. Fully suspended conditions at the beginning of every test were established by using rotating blades, connected to a drilling machine. Pore-fluid pressures were measured with piezoresistive transmitters (Keller Series) at 5, 15, 25, and 45 cm above the impermeable bottom of the tube at predetermined time intervals, ranging from one measurement per second to one measurement per 5 s depending on the clay content. At the end of each test, the final height H_w of the water column was measured and the pressure conditions were assumed to be hydrostatic. The diffusion coefficient D was determined by iterating between the measured and predicted excess fluid pressure (for details and equations we would like to refer to Major (2000)). The porosity of the beds with varying clay fractions was determined by inserting the different bed mixtures in soil sample rings, saturating the sediment in the rings with water, weighing the saturated samples, and comparing that to the dry weight of the sample.

The porosity of the initial beds in the flume is larger than the porosity of the dry unconsolidated material (random loose packing) and dry consolidated material (random close packing), which are, respectively, 0.34 and 0.27 for beds without clay. In the initial beds, the apparent cohesion caused by water in the unsaturated bed results in a larger porosity. Saturation of our bed mixtures occurs around a water volume fraction equal to the dry porosity, equal to a mass fraction of 14%–20% for a bed without clay. In our experiments, a water mass fraction >15% led to saturation and denser packing of the sediment during mixing. The effect of this dense packing is increased intergranular contact and higher resistance against erosion, and caused different behavior above the saturation threshold. These bed conditions were therefore excluded from the present analysis but are included in the online data set.

3. Results

3.1. General Erosion Trends for Varying Clay and Water Content of the Bed

The net-change patterns for the different experiments clearly show the significant influence of the clay and water content of the bed for debris-flow erosion (Figure 2, Figure S2 in Supporting Information S1). When the clay fraction of the bed is increased from 0 to 0.04, erosion increases (Figure 2a) with increasing scour at the upstream part of the erodible bed. Under a further increase of the clay fraction, from 0.06 to 0.1, erosion slowly ceases and becomes more homogeneous over the length of the bed (Figure 2a). The above-described trend is valid for all three flume angles under which experiments have been conducted (Figure 2b). However, under a flume angle of 28° , we observe a muted response in net change and erosion pattern, with less erosion but also less deposition. Under this flume angle, viscous forces dominate within the debris flow, in contrast to the dominance of frictional forces under higher flume angles. We hypothesize that this difference in flow characteristics under a flume angle of 28° explains the less pronounced erosion and deposition.

Under an increasing water fraction, up to 0.13, net change stays stable and net deposition occurs (Figure 2b). A further increase in water fraction results in a dramatic increase in net erosion under flume angles of 31° and 34° (Figure 2b). The spatial erosion patterns under different bed water contents are comparable, with scour at the top of the erodible bed and deposition on the lower half.

3.2. Diffusivity and Porosity for Different Clay Fractions of the Bed

To quantify how effectively interstitial fluid and pore pressure travel through the bed under varying clay content, the diffusivity and porosity of those different bed compositions were determined. With an increase in clay fraction, the diffusivity and porosity of the bed decrease exponentially (Figures 2c-2d). The exponential decrease in diffusivity and porosity as a function of the clay fraction shows that clay fills up the pore spaces and decreases the flow of interstitial fluid and the transfer of pore pressure through the bed.

3.3. Pore Pressure in the Bed Under Varying Bed Compositions

To study if and how changing pore pressures in the bed influence erosion during our experiments, we explore the temporal pattern of pore pressure relative to the initial conditions in the bed for six key bed conditions. In most experiments, we observe a decrease in relative pore pressure at initial debris flow impact (Figure 3), followed by an increase of 100–400 Pa, depending on the bed composition. This pressure is lower than the maximum normal





Figure 2. Overview of net change (cm³) under different (a) clay fractions (dry bulk mass fraction) and different (b) water fractions (fraction of total mass) in the erodible bed, as well as results of the diffusivity (c) and porosity (d) tests for varying clay fractions (fraction of dry weight) with exponential trend lines. For the first two panels (a, b), the different colors of the data points indicate the flume angle under which the experiment was conducted. Note that a negative net change means more sediment was eroded in the flume than was deposited, and vice versa for a positive net change.

force (based on flow depth) exerted by the debris flows on the bed, which on average ranges between 500 and 600 Pa, depending on the angle of the flume.

An increase in clay fraction has three notable effects on the pore pressure. First, the lowering of the pressure at flow-front arrival disappears (Figures 3a–3c). Second, from no clay to a clay fraction of 0.04, the maximum pore pressure becomes higher and dissipation of the increased pressure becomes slower (from 7.5 to 12 s at a clay fraction of 0.04, Figures 3a and 3b). Third, under the highest clay fraction, the change in pore pressure is significantly smaller, and the response is slow (Figure 3c).

Increasing the water content of the bed from 0.1 to 0.13 leads to an increase in the maximum pore pressure and a decrease in the initial pressure draw-down (Figures 3d and 3e). Under these conditions, we also observe the establishment of a new pressure equilibrium (Figure 3e, flattening of the blue line after 5 s). Under high water fractions of the bed (0.14), the increase in pore pressure is more rapid after flow-front arrival (Figure 3f), and the response in pore pressure is more chaotic. The latter could be explained by severe erosion around the sensors.

4. Discussion

4.1. Effects of Water and Clay in the Bed on Erosion

Our experiments show that bed composition strongly controls the magnitude of erosion by debris flows. The results of our experiments with varying bed water content are in agreement with earlier experimental results (Iverson et al., 2011; Reid et al., 2011) and field studies (de Haas et al., 2022; McCoy et al., 2012) that show that an increase in bed water content enhances erosion by debris flows. The studies by de Haas et al. (2022), Iverson et al. (2011), McCoy et al. (2012), and Reid et al. (2011) show a linear response in erosion magnitude to bed water content, whereas, in our experiments, erosion exponentially increases between a water fraction of 0.13 and 0.15. The difference between our and previous studies already hints at the possible importance of fines in the bed. In our water content experiments, fines were absent in the bed mixture, whereas fines were present in the bed in the



10.1029/2023GL103294



Figure 3. Relative pore-fluid pressure in the erodible bed measured by two pressure sensors (P1 and P2) installed at different depths during six representative experiments conducted at a flume angle of 34°. Flow depth of the debris flow overriding the bed is plotted in black. Panels (a–c) show results from experiments with increasing clay fractions (experiment 181, 153, and 157 respectively). Panels (d–f) show results from experiments with increasing water fractions (experiment 180, 201, and 187 respectively). Note that the pore pressure displayed is relative to the initial pressure created by the water in the protruding tubes. For pore pressures uncorrected for initial conditions see Figure S3 in Supporting Information S1. P2 records a higher pressure, explained by its lower position in the bed and thus the larger water column that can exist above it. Other differences between the data from the sensors can be explained by small heterogeneities in the bed, caused by either non-perfect mixing of the sediment, slight differences in packing during bed insertion, or other simple stochastics.

experiments by Iverson et al. (2011) and Reid et al. (2011), similar to natural settings. Due to the lack of fines in some of our experiments, drainage of water in the bed is unhindered (up to a water content of 0.13) and the increase in pore pressure is limited when a debris flow overrides the bed (see Figures 3d and 4a). In addition, the unhindered draining of fluid from a debris flow into the bed decreases the momentum and velocity of the debris flow (as shown by Iverson et al. (2011), Reid et al. (2011), and Roelofs et al. (2022)), further limiting the amount of erosion (see Figures 2b and 2c). Above a water content of 0.13 en-masse failure occurs (Figures S2c and S2d in Supporting Information S1) as the bed becomes saturated when overridden by the debris flow.

Our experiments with varying bed clay fractions further illustrate the significant effects of fines in the substrate for debris-flow erosion magnitude and processes (see Figure 2a). A small increase in the bed clay fraction (up to a dry weight fraction of 0.04) increases erosion. The clay in the bed decreases its diffusivity exponentially (Figures 2c and 4b), which decreases the ease at which fluids drain through the bed and increases the pore pressure in the bed directly underneath the debris flow (conceptualized in Figure 4b). This causes local undrained loading conditions, enhanced bed pore pressures, and erosion aided by liquefaction of the top of the bed related to an increase in water fraction of the bed (also discussed by e.g., Berger et al. (2011), Iverson (2012), Iverson et al. (2011), Major (2000), McCoy et al. (2012), and Sassa and hui Wang (2005)). In this case, the debris flow loses little momentum, which further enhances erosion (Iverson et al., 2011; Roelofs et al., 2022). The increased pore pressure decreases intergranular friction (Iverson, 1997), promoting erosion of the bed sediments by shear and impact forces. We hypothesize that the clay-related effects on erosion described above will be amplified under higher bed water content, in line with the findings of Iverson et al. (2011) and Reid et al. (2011).

A further increase of the bed clay fraction, beyond a fraction of 0.04, results in very limited erosion (conceptualized in Figure 4c). Under these conditions, we hypothesize that undrained loading still occurs, but that the infilling of the pore spaces by clay particles alters the response of the bed to shear, which becomes more dominated by dilation. This should result in decreasing pore pressure at debris-flow arrival, which we observe in one of the pore pressure sensors but not in both (Figure 3c). We expect that a small amount of compression still occurs as a result of the normal force exerted on the bed by the debris flow, which mutes the dilation response. Furthermore,



Figure 4. Schematic representation of the non-linear effects of bed clay content on erosion by debris flows. Without clay (a), interstitial fluid from the debris flow can drain unhindered through the bed and erosion occurs as limited scour due to shear and impact forces imposed on the bed. With the optimal amount of clay (b), in our experiments 2%–4% of the dry bulk mass of the bed, transfer of pore pressure in the bed occurs but is hindered by clay particles decreasing the diffusivity of the bed. Therefore, undrained loading occurs, and erosion is increased due to liquefaction of the top layer of the bed. With very high clay fractions in the bed (c), the soil is relatively more compacted due to the clay particles filling up the pore space and the behavior of the soil becomes more dominated by dilation. Therefore, erosion is limited.

the severe decrease in diffusivity under high bed clay fractions (see Figure S2c in Supporting Information S1) will hamper the transfer of water from the debris flow into the bed.

Our results thus show that a small increase in clay fraction (0%–4%) of the bed, while keeping the sand porosity roughly similar, can have a dramatic impact on erosion magnitude and thus volume growth and hazard potential. We do want to highlight that under different clay fractions in our experiments the total porosity decreased. We, therefore, cannot draw conclusions on the erodibility of bed mixtures of similar porosities with varying amounts of clay. It is of interest to note that the observed fines content in natural debris flows is consistent with our tested parameter space. Bulk fraction of fines in real-life debris flows and their deposits ranging from 2% to 20% have been reported (Ni et al., 2011; Phillips & Davies, 1991; Remaître et al., 2005; Yong et al., 2013). These deposits also form the unconsolidated beds in debris-flow gullies and variations in clay content may therefore help explain the widely contrasting erosion rates and magnitudes we observe in the field (e.g., de Haas et al., 2022; Hungr et al., 2005; Santi et al., 2008).

4.2. The Bed, the Flow, or Both?

Our results show that the composition of the bed can have a significant impact on both the amount of erosion caused by an overriding debris flow and the relative importance of contrasting processes. Previous experimental work has also shown that the composition of the debris flow affects erosion magnitude and processes (De Haas & Woerkom, 2016; Egashira et al., 2001; Fagents & Baloga, 2006; Hungr et al., 2005; Roelofs et al., 2022). Combining these results, we can state that debris-flow erosion is significantly affected by the abundance of water and clay in both the debris flow and the erodible bed. The importance of shear and impact forces on debris-flow erosion has long been recognized (Berger et al., 2011; de Haas et al., 2022; Frank et al., 2015; Hsu et al., 2008; Hungr et al., 2005; Mangeney et al., 2007; Roelofs et al., 2022; Stock & Dietrich, 2006; Takahashi, 1978, 1981),

as well as the importance of pore pressures for debris-flow dynamics (e.g., Costa, 1984; Iverson, 1997; Major & Iverson, 1999; McCoy et al., 2010). However, the influence that clay has on the relative importance of different erosion forces has been overlooked. This study and Roelofs et al. (2022) show that the effects of bed and debris flow composition, and especially clay and water content of both the unconsolidated bed and the debris flow itself, should be accounted for. The presence of water and fines directly affects the mobility and momentum of the debris flow (De Haas & Woerkom, 2016; Iverson et al., 2011; Roelofs et al., 2022), draining conditions (as also discussed by Roelofs et al., 2022), the effectiveness of pore pressure transfer, and the occurrence of liquefaction. Whereby the presence of clay can also influence bed porosity (as in this study) as well as the contractive or dilative behavior of the sediment.

Our results show that the composition of the soil can have a large but complex effect on debris-flow erosion and thus highlight the importance of incorporating bed composition effects in debris-flow erosion models. In current debris-flow erosion models, erosion is predicted based on the forces exerted on the bed by the debris flow (e.g., Chen & Zhang, 2015; Frank et al., 2017; Iverson, 2012; Iverson & Ouyang, 2015; Pudasaini & Fischer, 2020). Soil composition is at best incorporated as an erodibility factor (Baggio et al., 2021; Chen & Zhang, 2015; Frank et al., 2017; Gregoretti et al., 2019). We advocate that for accurate erosion prediction among different catchments, where calibration is not always possible, an erosion model in which the erodibility of the soil is described in a physics-based manner is needed.

4.3. How the Small Particles Matter—Lab Versus Field

Many studies have shown that small-scale debris flows in laboratory flumes can be used to study natural debrisflow behavior, and depositional and erosional mechanisms (De Haas et al., 2015; De Haas & Woerkom, 2016; Egashira et al., 2001; Iverson et al., 2011; Roelofs et al., 2022; Zheng et al., 2021). In our specific case, the erosion trends observed in our experiments clearly link to physical processes and parameters that affect debrisflow erosion in the field, that is, diffusivity of the bed, pore pressures, (un)drained loading conditions, contractive or dilative behavior of the sediment, and bed shear strength. Therefore, we believe that the trends in our data, related to clay and water content of the bed, are of relevance to the field.

However, scale effects cannot be fully neglected. In our study, special attention should be given to the reduced effects of fluid pore pressure in lab-scale debris flows (Iverson, 1997; Iverson & Denlinger, 2001; Iverson et al., 2010). The ability of a flow to retain excess fluid pressure increases quadratically over increasing flow depth, which significantly affects debris flow dynamics (Iverson & Denlinger, 2001) and possibly erosion. In our experiments, the recorded pore pressures in the bed did not rise above the normal force exerted by the debris flows, opposing observations from the field (McArdell et al., 2007; McCoy et al., 2012), larger-scale debris-flow experiments (Iverson et al., 2011), and debris-flow experiments in centrifuges (Bowman et al., 2010). Despite this discrepancy, the bed pore pressures are clearly influenced by changes in the water and clay content of the bed in our experiments, and there is no physical argument for why this would be different on a larger scale. However, it is likely that the trends we observe related to clay and water content of the bed might shift slightly in response to larger debris flows.

5. Conclusions

We studied the effects of bed composition on debris-flow erosion magnitude and processes by performing experiments in a flume with an erodible, unconsolidated bed. We tested the effects of the water and the clay content of the bed, while keeping the composition and volume of the debris flow constant. With data from DEMs, we quantified net change, and with data from pore pressure sensors and additional diffusivity and porosity tests, we identified the forces and processes working on and in the bed.

The results from our experiments show that the water and clay content of the bed influence erosion magnitude by affecting pore pressures in the bed, loading conditions, and dilative/contractive behavior. In our experiments, an optimum exists for maximum erosion under a specific clay content (4% of the dry bulk mass). Under this optimum clay fraction, drainage in the bed is partly hindered, resulting in undrained loading, elevated bed pore pressures, and possibly liquefaction of the top of the bed. Together, these reduce inter-particle friction and promote erosion. An increase in bed water content increases debris-flow erosion in our experiments by filling up pore space with water, resulting in elevated bed pore pressures when the debris flow overrides the bed. When the bed is close to saturation, this causes en-masse failure.

19448007, 2023, 14, Downlo

viley.com/doi/10.1029/2023GL103294 by Utrecht University Library

, Wiley Online Library on [11/10/2023]

. See

From our results, we can infer that small changes in hydrological and geological settings of a catchment may significantly impact debris-flow erosion, as small changes in soil moisture and the grain-size distribution of the sediment can lead to significant changes in final debris-flow volume and hazard potential. In addition, a changing climate and resulting environmental change, such as altered precipitation intensity, retreating glaciers, melting permafrost, and changing wildfire occurrence, influence the hydrological settings of the catchment as well as the availability and grain-size distribution of sediments. Therefore, understanding and incorporating bed effects in a more physics-based manner in debris-flow erosion modeling is important for current and future hazard prediction as well as for anticipating longer-term morphological change. Despite the importance of incorporating these effects for accurate predictions, we acknowledge that the data necessary to do so are difficult to obtain and not available for the vast majority of catchments. Therefore, effort should be made to test the relevance of our results in the field and assess if relatively easily obtainable predictors can be used to estimate bed erodibility (e.g., catchment lithology).

Data Availability Statement

DEM's and raw data from the pore pressure sensors are available via Yoda (online repository of Utrecht University). The data and an instruction on how we processed the raw data can be found under this link: https://public.yoda.uu.nl/geo/UU01/Y0RH2E.html. DOI: 10.24416/UU01-Y0RH2E.

References

- Abancó, C., & Hürlimann, M. (2014). Estimate of the debris-flow entrainment using field and topographical data. *Natural Hazards*, 71(1), 363–383. https://doi.org/10.1007/s11069-013-0930-5
- Baggio, T., Mergili, M., & D'Agostino, V. (2021). Advances in the simulation of debris flow erosion: The case study of the Rio Gere (Italy) event of the 4th August 2017. Geomorphology, 381, 107664. https://doi.org/10.1016/j.geomorph.2021.107664
- Bagnold, R. A. (1954). Experiments on a gravity-free dispersion of large solid spheres in a Newtonian fluid under shear. Proceedings of the Royal Society of London. Series A. Mathematical and Physical Sciences, 225(1160), 49–63. https://doi.org/10.1098/rspa.1954.0186
- Beguería, S., Van Asch, T. W., Malet, J.-P., & Gröndahl, S. (2009). A GIS-based numerical model for simulating the kinematics of mud and debris flows over complex terrain. *Natural Hazards and Earth System Sciences*, 9(6), 1897–1909. https://doi.org/10.5194/nhess-9-1897-2009
- Berger, C., McArdell, B., & Schlunegger, F. (2011). Direct measurement of channel erosion by debris flows, Illgraben, Switzerland. Journal of Geophysical Research, 116(F1). https://doi.org/10.1029/2010JF001722
- Blair, T. C., & McPherson, J. G. (1994). Alluvial fans and their natural distinction from rivers based on morphology, hydraulic processes, sedimentary processes, and facies assemblages. *Journal of Sedimentary Research*, 64(3a), 450–489. https://doi.org/10.1306/ D4267DDE-2B26-11D7-8648000102C1865D
- Bowman, E. T., Laue, J., Imre, B., & Springman, S. M. (2010). Experimental modelling of debris flow behaviour using a geotechnical centrifuge. *Canadian Geotechnical Journal*, 47(7), 742–762. https://doi.org/10.1139/T09-141
- Cannon, S. H., & DeGraff, J. (2009). The increasing wildfire and post-fire debris-flow threat in western USA, and implications for consequences of climate change. In K. Sassa & P. Canuti (Eds.), Landslides Disaster risk reduction (pp. 177–190). Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-540-69970-5_9
- Cavalli, M., & Marchi, L. (2008). Characterisation of the surface morphology of an alpine alluvial fan using airborne lidar. *Natural Hazards and Earth System Sciences*, 8(2), 323–333. https://doi.org/10.5194/nhess-8-323-2008
- Chen, H., & Zhang, L. M. (2015). Edda 1.0: Integrated simulation of debris flow erosion, deposition and property changes. Geoscientific Model Development, 8(3), 829–844. https://doi.org/10.5194/gmd-8-829-2015
- Conway, S., Decaulne, A., Balme, M., Murray, J., & Towner, M. (2010). A new approach to estimating hazard posed by debris flows in the Westfjords of Iceland. *Geomorphology*, 114(4), 556–572. https://doi.org/10.1016/j.geomorph.2009.08.015
- Costa, J. E. (1984). Physical geomorphology of debris flows. In Developments and applications of geomorphology (pp. 268–317). Springer.
- de Haas, T., Åberg, A. S., Walter, F., & Zhang, Z. (2021). Deciphering seismic and normal-force fluctuation signatures of debris flows: An experimental assessment of effects of flow composition and dynamics. *Earth Surface Processes and Landforms*, 46(11), 2195–2210. https:// doi.org/10.1002/esp.5168
- De Haas, T., Braat, L., Leuven, J. R., Lokhorst, I. R., & Kleinhans, M. G. (2015). Effects of debris flow composition on runout, depositional mechanisms, and deposit morphology in laboratory experiments. *Journal of Geophysical Research: Earth Surface*, 120(9), 1949–1972. https:// doi.org/10.1002/2015JF003525
- de Haas, T., Densmore, A., Stoffel, M., Suwa, H., Imaizumi, F., Ballesteros-Cánovas, J., & Wasklewicz, T. (2018). Avulsions and the spatio-temporal evolution of debris-flow fans. *Earth-Science Reviews*, 177, 53–75. https://doi.org/10.1016/j.earscirev.2017.11.007
- de Haas, T., & Densmore, A. L. (2019). Debris-flow volume quantile prediction from catchment morphometry. *Geology*, 47(8), 791–794. https:// doi.org/10.1130/g45950.1
- de Haas, T., McArdell, B. W., Nijland, W., Åberg, A. S., Hirschberg, J., & Huguenin, P. (2022). Flow and bed conditions jointly control debrisflow erosion and bulking. *Geophysical Research Letters*, 49(10), e2021GL097611. https://doi.org/10.1029/2021GL097611
- De Haas, T., Nijland, W., De Jong, S., & McArdell, B. (2020). How memory effects, check dams, and channel geometry control erosion and deposition by debris flows. *Scientific Reports*, 10(1), 1–8. https://doi.org/10.1038/s41598-020-71016-8
- De Haas, T., & van Woerkom, T. (2016). Bed scour by debris flows: Experimental investigation of effects of debris-flow composition. *Earth Surface Processes and Landforms*, 41(13), 1951–1966. https://doi.org/10.1002/esp.3963
- De Haas, T., Ventra, D., Carbonneau, P. E., & Kleinhans, M. G. (2014). Debris-flow dominance of alluvial fans masked by runoff reworking and weathering. *Geomorphology*, 217, 165–181. https://doi.org/10.1016/j.geomorph.2014.04.028
- Dowling, C. A., & Santi, P. M. (2014). Debris flows and their toll on human life: A global analysis of debris-flow fatalities from 1950 to 2011. *Natural Hazards*, 71(1), 203–227. https://doi.org/10.1007/s11069-013-0907-4

Acknowledgments

This work was supported by the Dutch Research Council (NWO) (Grant 0.16. Veni.192.001 and Grant OCENW. KLEIN.495 to TdH). The authors gratefully acknowledge the help of Arian van Eijk, Bas van Dam, Marcel van Maarseveen, Henk Markies en Mark Eijkelboom in designing and constructing the flume and measurement set-ups, as well as their assistance during the experiments. The authors would also like to thank Maarten Kleinhans for his contribution to improving the first draft of this manuscript and contributing to our understanding of some of the basic concepts around the influence of moisture content to sand packing and porosity. We would like to thank Dr. Elisabeth Bowman and one anonymous reviewer for their constructive and thorough response to the first version of this manuscript.

- Egashira, S., Honda, N., & Itoh, T. (2001). Experimental study on the entrainment of bed material into debris flow. *Physics and Chemistry of the Earth, Part C: Solar, Terrestrial & Planetary Science*, 26(9), 645–650. https://doi.org/10.1016/s1464-1917(01)00062-9
- Ékes, C., & Friele, P. (2003). Sedimentary architecture and post-glacial evolution of Cheekye Fan, southwestern British Columbia, Canada. Geological Society, London, Special Publications, 211(1), 87–98. https://doi.org/10.1144/GSL.SP.2001.211.01.08
- Fagents, S. A., & Baloga, S. M. (2006). Toward a model for the bulking and debulking of lahars. *Journal of Geophysical Research*, 111(B10), B10201. https://doi.org/10.1029/2005jb003986
- Frank, F., McArdell, B., Huggel, C., & Vieli, A. (2015). The importance of erosion for debris flow runout modelling from applications to the Swiss Alps. Natural Hazards and Earth System Sciences Discussions, 3(49), 2379–2417. https://doi.org/10.5194/nhessd-3-2379-2015
- Frank, F., McArdell, B. W., Oggier, N., Baer, P., Christen, M., & Vieli, A. (2017). Debris-flow modeling at Meretschibach and Bondasca catchments, Switzerland: Sensitivity testing of field-data-based entrainment model. *Natural Hazards and Earth System Sciences*, 17(5), 801–815. https://doi.org/10.5194/nhess-17-801-2017

Giraud, R. E. (2005). Guidelines for the geologic evaluation of debris-flow hazards on alluvial fans in Utah. Utah Geological Survey, 5(6).

- Gregoretti, C., Stancanelli, L. M., Bernard, M., Boreggio, M., Degetto, M., & Lanzoni, S. (2019). Relevance of erosion processes when modelling in-channel gravel debris flows for efficient hazard assessment. *Journal of Hydrology*, 568, 575–591. https://doi.org/10.1016/j. jhydrol.2018.10.001
- Han, Z., Chen, G., Li, Y., Zhang, H., & He, Y. (2016). Elementary analysis on the bed-sediment entrainment by debris flow and its application using the TopFlowDF model. *Geomatics, Natural Hazards and Risk*, 7(2), 764–785. https://doi.org/10.1080/19475705.2014.966868
- Hsu, L., Dietrich, W. E., & Sklar, L. S. (2008). Experimental study of bedrock erosion by granular flows. *Journal of Geophysical Research*, 113(F2), F02001. https://doi.org/10.1029/2007JF000778
- Hungr, O., McDougall, S., & Bovis, M. (2005). Entrainment of material by debris flows. In *Debris flow hazards and related phenomena* (pp. 135–158). Springer Berlin Heidelberg.
- Iverson, R. M. (1997). The physics of debris flows. Reviews of Geophysics, 35(3), 245-296. https://doi.org/10.1029/97RG00426
- Iverson, R. M. (2012). Elementary theory of bed-sediment entrainment by debris flows and avalanches. Journal of Geophysical Research, 117(F3). https://doi.org/10.1029/2011jf002189
- Iverson, R. M., & Denlinger, R. P. (2001). Flow of variably fluidized granular masses across three-dimensional terrain: 1. Coulomb mixture theory. Journal of Geophysical Research, 106(B1), 537–552. https://doi.org/10.1029/2000jb900329
- Iverson, R. M., Logan, M., LaHusen, R. G., & Berti, M. (2010). The perfect debris flow? Aggregated results from 28 large-scale experiments. Journal of Geophysical Research, 115(F3), 1–17, F03005. https://doi.org/10.1029/2009JF001514
- Iverson, R. M., & Ouyang, C. (2015). Entrainment of bed material by earth-surface mass flows: Review and reformulation of depth-integrated theory. *Reviews of Geophysics*, 53(1), 27–58. https://doi.org/10.1002/2013rg000447
- Iverson, R. M., Reid, M. E., Logan, M., LaHusen, R. G., Godt, J. W., & Griswold, J. P. (2011). Positive feedback and momentum growth during debris-flow entrainment of wet bed sediment. *Nature Geoscience*, 4(2), 116–121. https://doi.org/10.1038/ngeo1040
- Jakob, M. (2005). Debris-flow hazard analysis. In Debris-flow hazards and related phenomena (pp. 411-443). Springer. https://doi. org/10.1007/3-540-27129-5_17
- Li, P., Wang, J., Hu, K., & Wang, F. (2020). Experimental study of debris-flow entrainment over stepped-gradient beds incorporating bed sediment porosity. *Engineering Geology*, 274, 105708. https://doi.org/10.1016/j.enggeo.2020.105708
- Lugon, R., & Stoffel, M. (2010). Rock-glacier dynamics and magnitude-frequency relations of debris flows in a high-elevation watershed: Ritigraben, Swiss Alps. *Global and Planetary Change*, 73(3), 202–210. https://doi.org/10.1016/j.gloplacha.2010.06.004
- Luna, B. Q., Remaître, A., van Asch, T., Malet, J.-P., & van Westen, C. (2012). Analysis of debris flow behavior with a one dimensional run-out model incorporating entrainment. *Engineering Geology*, 128, 63–75. (Integration of Technologies for Landslide Monitoring and Quantitative Hazard Assessment). https://doi.org/10.1016/j.enggeo.2011.04.007
- Major, J. J. (2000). Gravity-driven consolidation of granular slurries: Implications for debris-flow deposition and deposit characteristics. Journal of Sedimentary Research, 70(1), 64–83. https://doi.org/10.1306/2dc408ff-0e47-11d7-8643000102c1865d
- Major, J. J., & Iverson, R. M. (1999). Debris-flow deposition: Effects of pore-fluid pressure and friction concentrated at flow margins. *Geological Society of America Bulletin*, 111(10), 1424–1434. https://doi.org/10.1130/0016-7606(1999)111<1424:dfdeop>2.3.co;2
- Mangeney, A., Bouchut, F., Thomas, N., Vilotte, J.-P., & Bristeau, M. (2007). Numerical modeling of self-channeling granular flows and of their levee-channel deposits. *Journal of Geophysical Research*, 112(F2), F02017. https://doi.org/10.1029/2006JF000469
- McArdell, B. W., Bartelt, P., & Kowalski, J. (2007). Field observations of basal forces and fluid pore pressure in a debris flow. *Geophysical Research Letters*, 34(7), L07406. https://doi.org/10.1029/2006GL029183
- McCoy, S. W., Kean, J. W., Coe, J. A., Staley, D. M., Wasklewicz, T. A., & Tucker, G. E. (2010). Evolution of a natural debris flow: In situ measurements of flow dynamics, video imagery, and terrestrial laser scanning. *Geology*, 38(8), 735–738. https://doi.org/10.1130/G30928.1
- McCoy, S. W., Kean, J. W., Coe, J. A., Tucker, G., Staley, D. M., & Wasklewicz, T. (2012). Sediment entrainment by debris flows: In situ measurements from the headwaters of a steep catchment. *Journal of Geophysical Research*, 117, F03016. https://doi.org/10.1029/2011JF002278
- Navratil, O., Liébault, F., Bellot, H., Travaglini, E., Theule, J., Chambon, G., & Laigle, D. (2013). High-frequency monitoring of debris-flow propagation along the Réal Torrent, Southern French Prealps. *Geomorphology*, 201, 157–171. https://doi.org/10.1016/j.geomorph.2013.06.017
- Ni, H., Zheng, W., Liu, X., & Gao, Y. (2011). Fractal-statistical analysis of grain-size distributions of debris-flow deposits and its geological implications. *Landslides*, 8(2), 253–259. https://doi.org/10.1007/s10346-010-0240-x
- Phillips, C. J., & Davies, T. R. (1991). Determining rheological parameters of debris flow material. *Geomorphology*, 4(2), 101–110. https://doi.org/10.1016/0169-555X(91)90022-3
- Pudasaini, S. P., & Fischer, J.-T. (2020). A mechanical erosion model for two-phase mass flows. International Journal of Multiphase Flow, 132, 103416. https://doi.org/10.1016/j.ijmultiphaseflow.2020.103416
- Rebetez, M., Lugon, R., & Baeriswyl, P.-A. (1997). Climatic change and debris flows in high mountain regions: The case study of the Ritigraben Torrent (Swiss Alps). In *Climatic change at high elevation sites* (pp. 139–157). Springer. https://doi.org/10.1007/978-94-015-8905-5_8
- Reid, M. E., Iverson, R. M., Logan, M., Lahusen, R. G., Godt, J. W., & Griswold, J. P. (2011). Entrainment of bed sediment by debris flows: Results from large-scale experiments. *Italian Journal of Engineering Geology and Environment*, 367–374. https://doi.org/10.4408/ IJEGE.2011-03.B-042
- Remaître, A., Malet, J.-P., & Maquaire, O. (2005). Morphology and sedimentology of a complex debris flow in a clay-shale basin. Earth Surface Processes and Landforms, 30(3), 339–348. https://doi.org/10.1002/esp.1161
- Rickenmann, D. (1999). Empirical relationships for debris flows. Natural Hazards, 19(1), 47–77. https://doi.org/10.1023/a:1008064220727
- Rickenmann, D. (2005). Runout prediction methods. In *Debris-flow hazards and related phenomena* (pp. 305–324). Springer Berlin Heidelberg. Roelofs, L., Colucci, P., & de Haas, T. (2022). How debris-flow composition affects bed erosion quantity and mechanisms - An experimental assessment. *Earth Surface Processes and Landforms*, 47(8), 2151–2169. https://doi.org/10.1002/esp.5369

- Santi, P. M., dewolfe, V. G., Higgins, J. D., Cannon, S. H., & Gartner, J. E. (2008). Sources of debris flow material in burned areas. *Geomorphology*, 96(3–4), 310–321. https://doi.org/10.1016/j.geomorph.2007.02.022
- Sassa, K., & hui Wang, G. (2005). Mechanism of landslide-triggered debris flows: Liquefaction phenomena due to the undrained loading of torrent deposits. In *Debris-flow hazards and related phenomena* (pp. 81–104). Springer.
- Simoni, A., Bernard, M., Berti, M., Boreggio, M., Lanzoni, S., Stancanelli, L. M., & Gregoretti, C. (2020). Runoff-generated debris flows: Observation of initiation conditions and erosion–deposition dynamics along the channel at Cancia (Eastern Italian Alps). *Earth Surface Processes and Landforms*, 45(14), 3556–3571. https://doi.org/10.1002/esp.4981
- Stock, J. D., & Dietrich, W. E. (2003). Valley incision by debris flows: Evidence of a topographic signature. Water Resources Research, 39(4), 1089. https://doi.org/10.1029/2001WR001057
- Stock, J. D., & Dietrich, W. E. (2006). Erosion of steepland valleys by debris flows. Geological Society of America Bulletin, 118(9–10), 1125– 1148. https://doi.org/10.1130/B25902.1
- Stoffel, M., & Huggel, C. (2012). Effects of climate change on mass movements in mountain environments. *Progress in Physical Geography: Earth and Environment*, 36(3), 421–439. https://doi.org/10.1177/0309133312441010
- Stoffel, M., Mendlik, T., Schneuwly-Bollschweiler, M., & Gobiet, A. (2014). Possible impacts of climate change on debris-flow activity in the Swiss Alps. *Climatic Change*, 122(1), 141–155. https://doi.org/10.1007/s10584-013-0993-z
- Takahashi, T. (1978). Mechanical characteristics of debris flow. Journal of the Hydraulics Division, 104(8), 1153–1169. https://doi.org/10.1061/jyceaj.0005046
- Takahashi, T. (1981). Debris flow. Annual Review of Fluid Mechanics, 13(1), 57–77. https://doi.org/10.1146/annurev.fl.13.010181.000421 Wan, S., & Lei, T. C. (2009). A knowledge-based decision support system to analyze the debris-flow problems at Chen-Yu-Lan River, Taiwan.
- Knowledge-Based Systems, 22(8), 580–588. https://doi.org/10.1016/j.knosys.2009.07.008 Welsh, A., & Davies, T. (2011). Identification of alluvial fans susceptible to debris-flow hazards. *Landslides*, 8(2), 183–194. https://doi.
- wilford, D., Sakals, M., Innes, J., Sidle, R. C., & Bergerud, W. (2004). Recognition of debris flow, debris flow debris flow
- shed morphometrics. *Landslides*, 1(1), 61–66. https://doi.org/10.1007/s10346-003-0002-0
- Yong, L., Xiaojun, Z., Pengcheng, S., Yingde, K., & Jingjing, L. (2013). A scaling distribution for grain composition of debris flow. Geomorphology, 192, 30–42. https://doi.org/10.1016/j.geomorph.2013.03.015
- Zheng, H., Shi, Z., Yu, S., Fan, X., Hanley, K. J., & Feng, S. (2021). Erosion mechanisms of debris flow on the sediment bed. Water Resources Research, 57(12), e2021WR030707. https://doi.org/10.1029/2021WR030707
- Zhou, G. G., & Ng, C. W. (2010). Dimensional analysis of natural debris flows. Canadian Geotechnical Journal, 47(7), 719–729. https://doi. org/10.1139/T09-134
- Zou, Q., Cui, P., Jiang, H., Wang, J., Li, C., & Zhou, B. (2020). Analysis of regional river blocking by debris flows in response to climate change. *Science of the Total Environment*, 741, 140262. https://doi.org/10.1016/j.scitotenv.2020.140262